Author's response to reviews

Title: The influence of the cultural climate of the training environment on physicians' self-perception of competence and preparedness for practice.

Authors:

Jamiu O Busari (j.busari@atriummc.nl)
Fred D Muskiet (fdmuski@yahoo.com)
Eduard AA Verhagen (a.a.e.verhagen@bkk.umcg.nl)

Version: 7 Date: 1 November 2008

Author's response to reviews: see over
Response to editor’s comment

Title: The influence of the cultural climate of the training environment on physicians' self-perception of competence and preparedness for practice.

Version: 6 Date: 01 November, 2008

Peer review of your manuscript (above) is now complete, and we are delighted, in principle, to accept the manuscript for publication in BMC Medical Education.....However, before acceptance, we would request that you add some more context information to the Background section of the Abstract.

We have added more context information to the background as requested

You also need to move the statements regarding ethical approval and consent to the Methods section of the manuscript.

We have moved the statements regarding ethical approval and consent to the methods section as requested (pp6; line 11-14)

We would also request that you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points.

We have revised the formatting of the manuscript as requested